Author's response to reviews

Title: The changes in clot microstructure in patients with ischaemic stroke and the effects of therapeutic intervention: a prospective observational study

Authors:

Sophia N Stanford (S.N.Stanford@swansea.ac.uk)
Ahmed Sabra (drsabraa@yahoo.com)
Lindsay D'Silva (L.A.DSilva@swansea.ac.uk)
Matthew Lawrence (m.j.lawrence@swansea.ac.uk)
Roger HK Morris (KMorris@cardiffmet.ac.uk)
Sharon Storton (sharon.storton@wales.nhs.uk)
Martyn R Brown (M.R.Brown@swansea.ac.uk)
Vanessa J Evans (VanessaJane.Evans@wales.nhs.uk)
Karl Hawkins (K.M.Hawkins@swansea.ac.uk)
Phylip R Williams (P.R.Williams@swansea.ac.uk)
Simon J Davidson (S.Davidson@rbht.nhs.uk)
Mushtaq Wani (Mushtaq.Wani@wales.nhs.uk)
John F Potter (john.potter@uea.ac.uk)
Phillip A Evans (Phillip.Evans2@wales.nhs.uk)

Version: 4 Date: 4 February 2015

Author's response to reviews: see over
Dear Editor,

**RE: Title: The changes in clot microstructure in patients with ischaemic stroke and the effects of therapeutic intervention: a prospective observational study**

Thank you for accepting our manuscript for publication following submission of the revised draft.

We would like to thank the reviewers again for their time and valuable comments and thanks to Professor Suzuki for his attention to details. Now the repeated word has been removed and a revised manuscript uploaded.

We look forward to hearing from you in due course.

Yours sincerely,

P. A. Evans

Professor Phillip Adrian Evans
Professor of Emergency Medicine and Haemostasis
NISCHR Haemostasis Biomedical Research Unit
Accident & Emergency Department
Morriston Hospital
ABMU Health Board
Swansea
SA6 6NL
United Kingdom
E-mail: phillip.evans2@wales.nhs.uk
Tel: +44 1792 70 3418